Craig Chambers CSE 501 Reading Assignments

Reading Assignment, Weeks #8, #9, #10

For: Read: 2/22 Section 14.7, [Grove et al. 97] 2/24 2/26 skim Chapter 9, skim 8.2, read [Gibbons & Muchnick 86]

3/1 Section 10.1 (on interference graphs), Chapter 11, [Briggs et al. 94] 3/3 [Wall 86] 3/5 [Padua & Wolfe 86], Sections 21.3-21.5

3/8 Sections 13.1-13.5, 13.7, 13.Further Reading (about conservative collection) 3/10 Sections 15.1-15.3, 15.5; [Kranz et al. 86] 3/12 Section 14.1-14.6 (note that this chapter is pretty naive and section 14.3 is really dumb);

[Hölzle & Ungar 94], [Dean et al. 96]

Bibliography:

[Grove et al. 97] David Grove, Greg DeFouw, Jeffrey Dean, and Craig Chambers. Call Graph Construction in Object-Oriented Languages. In OOPSLA ’97.

[Gibbons & Muchnick 86] Phillip B. Gibbons and Steven S. Muchnick. Efficient Instruction Scheduling for a Pipelined Architecture. In PLDI ’86.

[Briggs et al. 94] Preston Briggs, Keith D. Cooper, and Linda Torczon. Improvements to Graph Coloring Register Allocation. In TOPLAS 16(3), May 1994.

[Wall 86] David W. Wall. Global Register Allocation at Link Time. In PLDI’86. [Padua & Wolfe 86] David A. Padua and Michael J. Wolfe. Advanced Compiler Optimizations for Supercomputers. In Communications of the ACM 29(12), December, 1986.

[Kranz et al. 86] David Kranz, Richard Kelsey, Jonathan Rees, Paul Hudak, James Philbin, and Norman Adams. ORBIT: An Optimizing Compiler for Scheme. In PLDI’86.

[Hölzle & Ungar 94] Urs Hölzle and David Ungar. Optimizing Dynamically Dispatched Calls with Run-Time Type Feedback. In PLDI ’94.

[Dean et al. 96] Jeffrey Dean, Greg DeFouw, David Grove, Vassily Litvinov, and Craig Chambers.

Vortex: An Optimizing Compiler for Object-Oriented Languages. In OOPSLA ’96.